

## CHANGES OF THE ORIGINAL MANUSCRIPT

We have checked through the manuscript and made the appropriate changes. Changes of individual words to clarify for contextual accuracy and better reader comprehension were highlighted by red colour.

## RESPONSE TO THE REVIEWER

Review 1#: A case report which may be publish according to editorial decision.

**Response:** Thank you for your valuable comment. We have modified the manuscript according to the editorial suggestion.

Review 2#: I read the manuscript named "Association of paraganglioma with thyroid carcinoma: a rare mimic of multiple endocrine tumors". (Manuscript NO: 45874) and my recommendations are as fallows. Title: It is accurately reflects the major topic and contents of the study. Abstract: Adequate, summarizing the topic. Cases report: Cases has been presented in detail. Topics has been discussed with all aspects. References: References are appropriate and updated. Figures are reflects the major findings of the study, and they are appropriately presented. This manuscript was well-written and documented. It is crucial that the author point out to rarely case report. This manuscript gives additional new knowledge to the literature. I think that this manuscript is suitable and worth to be published in the World Journal of Clinical Cases.

**Response:** Thank you for your valuable comment.

## RESPONSE TO THE PEER-REVIEWERS

*Please provide language certificate letter by professional English language editing companies (Classification of manuscript language quality evaluation is B).*

**Response:** We have provided and uploaded the language certification.

*Audio core tip: In order to attract readers to read your full-text article, we request that the author make an audio file describing your final core tip, it is necessary for final acceptance. Please refer to Instruction to authors on our website or attached Format for detailed information. The accepted formats are mp3 or wma.*

**Response:** Thank you for the suggestion. We have provided and uploaded the audio core tip file.

*Please provide the decomposable figure of Figures, whose parts are movable and editable. So you can put the original pictures in ppt and submit it in the system.*

**Response:** We have provided and uploaded the PPT.

*Please modify the manuscript according to the result of crosscheck, especially part 1.*

**Response:** We have modified the manuscript according to the result of the crosscheck. Changes of individual words were highlighted by red colour.

*Our policy for the Case Report manuscript requires the title to include the disease name, the number of cases, and the phrase "literature review". In addition, the description of the paper as a literature review must be supported by the manuscript's content. Please update the reference list and add references with accompanying textual content that will strengthen the manuscript as a literature review of the appropriate and up-to-date case-related information. Please note that if authors only add the words "literature review" to the title, but do not revise the textual content of the manuscript to provide a literature review, the manuscript will be rejected.*

**Response:** The title was changed as: "Concomitant of paraganglioma and

thyroid carcinoma: an unusual case report and literature review". The content of the review and a table have been added to the manuscript in the Discussion part.

*Please provide the author contributions. See the format in the attachment file-revision policies.*

**Response:** The information of authors` contribution was provided.

*Please provide the Corresponding author's name, title, and detailed address.*

**Response:** The corresponding author`s information was provided.

*The structured abstract should be at least 250 words. The abstract subsections will include background, case summary, and conclusion, written as: BACKGROUND (no more than 80 words) What does this case report add to the medical literature? Why did you write it up? CASE SUMMARY (no more than 150 words) What were the chief complaints, diagnoses, interventions, and outcomes? CONCLUSION (no more than 20 words) What is the main "take-away" lesson from this case?*

**Response:** We have rewritten the Abstract part.

*Please change the main body of the case report to the following format.*

**Response:** We have changed the main body of the case report as required.

*Please check and confirm that there are no repeated references !*

**Response:** There are no repeated references.

*Please add PubMed citation numbers (**PMID** NOT PMCID) and DOI citation to the reference list and **list all authors**. Please revise throughout. The author should provide the first page of the paper without PMID and DOI.*

**Response:** The PMID and DOI information has been added to the references.